Comments on Assignment #2

Question #1: You received 10 marks if you tried this question and 0 if you did not. See the solutions to check if your answer is correct.

Question #2: Many people lost marks again for failing to use the appropriate format for a modelling question. When determining your model, make sure you read the question carefully, the details in the wording of the problem will give you many clues as to how the program should look. When you think you have a ’good’ program describing your problem, look it over and 1) make sure it makes sense, and 2) make sure all the little details were taken care of. If you make ANY assumptions please state them as such.

Question #3: Eighty percent of the class did not read the question carefully enough and thus proved the wrong thing in part a. You were not supposed to prove the feasible region is convex. Everyone should look at the solution for this question.

Question #4: You received 10 marks if you tried this question and 0 if you did not. See the solutions to check if your answer is correct.